Author's response to reviews

Title: The functional status and well being of people with myalgic encephalomyelitis/chronic fatigue syndrome and their carers

Authors:

Luis C Nacul (luis.nacul@lshtm.ac.uk)
Eliana M Lacerda (eliana.lacerda@lshtm.ac.uk)
Peter Campion (p.d.campion@hull.ac.uk)
Derek Pheby (derekpheby@btinternet.com)
Maria de L Drachler (jcdc.leite@googlemail.com)
Jose C Leite (jcdc.leite@googlemail.com)
Fiona Poland (F.Poland@uea.ac.uk)
Amanda Howe (Amanda.Howe@uea.ac.uk)
Shagufta Fayyaz (Shagufta.Fayyaz@lshtm.ac.uk)
Mariam Molokhia (Mariam.Molokhia@kcl.ac.uk)

Version: 5 Date: 11 May 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

We are pleased to enclose the final version of the manuscript, addressing the required changes in its format, as shown below.

Major revisions
1. Authors list: We ensured consistency in the submission system and in the manuscript
2. Abstract: changed in the submission system, which is now consistent to the manuscript.
3. Additional files uploaded are now referred accordingly in the manuscript, as required by the reviewers.

Minor revisions
1. Highlighting/tracking were removed
2. References – the references now show the list of authors accordingly, with up to 30 authors listed.
3. Tables: the vertical lines were taken off from the tables.
4. Figures: the figures were changed to comply with the requirements.
5. Typography: a couple of minor errors were corrected in the manuscript.

We look forward to seeing the paper published at BMC Public Health.

Yours sincerely,
Dr Luis Nacul (on behalf of the authors)